

Geophysical Research Letters

RESEARCH LETTER

10.1029/2020GL087478

Key Points:

- Drought and flood extremes in the Amazon River basin have increased in frequency during the last 40 years
- A new tree-ring chronology of Cedrela odorata confirms the dating and climate signal of a previous chronology for the eastern Amazon
- Multidecadal variability of tree-ring reconstructed rainfall suggests that recent eastern Amazon rainfall extremes may not be unprecedented

Correspondence to:

D. Granato-Souza, danigsbio@gmail.com

Citation:

Granato-Souza, D., Stahle, D. W., Torbenson, M. C. A., Howard, I. M., Barbosa, A. C., Feng, S., et al.. (2020). Multidecadal changes in wet season precipitation totals over the eastern Amazon. *Geophysical Research Letters*, 47, e2020GL087478. https://doi.org/ 10.1029/2020GL087478

Received 10 FEB 2020 Accepted 30 MAR 2020 Accepted article online 9 APR 2020

Multidecadal Changes in Wet Season Precipitation Totals Over the Eastern Amazon

Daniela Granato-Souza¹, David W. Stahle¹, Max C.A. Torbenson², Ian M. Howard¹, Ana Carolina Barbosa³, Song Feng¹, Katia Fernandes^{1,4}, and Jochen Schöngart⁵

¹Department of Geosciences, University of Arkansas, Fayetteville, AR, USA, ²Civil, Environmental and Geodetic Engineering, Ohio State University, Columbus, OH, USA, ³Department of Forest Sciences, Federal University of Lavras, Lavras, Brazil, ⁴International Research Institute for Climate and Society, Columbia University, Palisades, NY, USA, ⁵Instituto Nacional de Pesquisas da Amazonia (INPA), Manaus, Brazil

Abstract Instrumental observations indicate that Amazon precipitation and streamflow extremes have increased during the last 40 years, possibly due to anthropogenic changes and natural variability. How unprecedented these changes might be is difficult to determine because some paleoclimatic, instrumental, and climate model simulations suggest that Amazonian precipitation and streamflow may be subject to multidecadal variability with return intervals longer than most direct observations. A new 258-year long tree-ring chronology of *Cedrela odorata* has been developed in the eastern Amazon and has been used to reconstruct wet season precipitation totals from 1759–2016. Reconstructed drought extremes are associated with significant sea surface temperature anomalies over the tropical Pacific and Atlantic Oceans. Strong multidecadal variance is identified in the reconstruction that may reflect a component of natural rainfall variability relevant to forest ecosystem dynamics and suggesting that recent hydroclimate changes over the eastern Amazon may not be unprecedented over the past 258 years.

Plain Language Summary Drought and flood extremes in the Amazon River basin have increased in frequency during the last 40 years, but how unprecedented these recent changes might be is difficult to determine because most instrumental records do not extend before 1950. A new tree-ring chronology of *Cedrela odorata* has been developed in the eastern equatorial Amazon, confirming the exact dating and climate signal of the single existing tree-ring chronology from the region. A tree-ring reconstruction of wet season precipitation based on this new chronology extends from 1759–2016 and documents strong multidecadal variability that impacted the frequency of reconstruction precipitation extremes over the past 258 years.

1. Introduction

The vast Amazon basin still preserves the largest tropical forests on Earth and plays a significant role in the general circulation of the atmosphere and the global energy balance (Nobre et al., 2016). Climate model simulations of hydroclimatic variability in the Amazon indicate that anthropogenic forcing may amplify the hydrologic cycle, resulting in more severe drought, extreme floods, and a decline in annual precipitation over most of the watershed (Feng & Fu, 2013). Some of these hydroclimatic changes may already be evident in observational records (Espinoza et al., 2014; Gloor et al., 2013; Marengo & Espinoza, 2016). Precipitation and streamflow extremes have increased in the Amazon during the last 40 years, possibly due to interactions between deforestation and climate (Khanna et al., 2017), intensification of the Walker Circulation (Barichivich et al., 2018), or global climate change (Nobre et al., 2016). How unprecedented these recent changes in extremes might be is difficult to determine because instrumental precipitation records have become widely available across the Amazon in only the last 50 years and high-resolution paleoclimate proxies are scarce.

The El Niño/Southern Oscillation (ENSO) strongly modulates the interannual variability of rainfall over Amazonia, with drought often developing during El Niño events (Marengo & Espinoza, 2016; Ronchail et al., 2002; Ropelewski & Halpert, 1987; Zeng, 1999). Lower-frequency variability may also modulate precipitation over the equatorial Amazon, and multidecadal regimes have been identified in speleothem proxies, instrumental observations, and climate model simulations over the basin (Ault et al., 2012; Chelliah & Bell, 2004; Fernandes et al., 2015; Labat et al., 2004; Marengo, 2004; Wang et al., 2011; Wang et al., 2017).

©2020. American Geophysical Union. All Rights Reserved.





Figure 1. The rainfall recording stations in the Amazon basin operating during each of 4 years are mapped from 1920–2000 (a–d; from ANA, Brazil, and in the Western Amazon from the Climatic Research Unit). Instrumental rainfall totals (mm) for Manaus are plotted for the wet (e; 1902–2016, October–June) and dry season (f; 1901–2015, July–September). These seasonal totals are not homogeneous when compared with Rio Negro maximum and minimum water levels (p < 0.05). The two *C. odorata* chronologies from the Rio Paru basin are plotted (g; RPA = red, RPB = black) along with the correlation between the chronologies (for 1786-2016 and 1851-2016) and the sample size each year for RPB.

Decadal episodes without drought or flood extremes are apparent in the Amazon River water level records at Manaus and Obidos from 1903-2015 (Barichivich et al., 2018), and in precipitation and terrestrial water storage indices from 1980-2016 (Chaudhari et al., 2019). Low-frequency variations are nevertheless difficult to document in the short, sparse, and often-interrupted instrumental measurements of precipitation in the Amazon that begin mostly in the mid- to late-20th century (e.g., Marengo, 2004; Garreaud et al., 2009). High-resolution proxy precipitation data could help document multidecadal variability in the preinstrumental period. The first moisture sensitive tree-ring chronologies of Cedrela odorata have recently been developed in the Amazon basin (Brienen et al., 2012; Granato-Souza et al., 2018), a significant accomplishment because most tropical tree species do not form consistent annual growth rings suitable for dendroclimatic analysis. The 231-year long Cedrela chronology from the eastern Amazon was used to reconstruct wet season rainfall from 1786-2016 (Granato-Souza et al., 2018).

A new 258-**year** long tree-ring chronology of *C. odorata* has been developed in the Rio Paru drainage basin of the eastern Amazon (Rio Paru-Baixo [RPB]). This chronology confirms the exact dating and precipitation response of the first *Cedrela* chronology developed from a separate location on the Rio Paru (i.e., Rio Paru-Alto; Granato-Souza et al., 2018). Positive and especially negative growth extremes in the new RPB *Cedrela* chronology are associated with above and below average precipitation at many instrumental rainfall stations across the Amazon basin. The RPB chronology is used to reconstruct wet season precipitation totals for the eastern equatorial Amazon from 1759–2016. The reconstruction is associated with ENSO extremes and exhibits significant multidecadal variability. Strong low frequency variability in reconstructed wet season precipitation suggests that the recent increase in high and low rainfall extremes may not be entirely unprecedented over the eastern Amazon basin in the context of natural variability.

2. Data and Methods

The new RPB chronology was developed from cross sections of *C. odorata* salvaged at a logging concession in the eastern Amazon (0.997522°S 53.2667°W) located 10 km to the southeast and 80 m lower in elevation than the site at Rio Paru-Alto (RPA; 0.97669°S 53.32586°W). The owners and employees of the logging firm CEMAL and the Environmental and Sustainability Agency in the state of Para facilitated the collection of cores and cross sections from legally harvested trees. All tree-ring specimens have been cataloged and permanently stored in the Tree Ring Laboratory of the Federal University of Lavras where they are available for further scientific research.

All cross sections were dried and highly polished to reveal the cellular anatomy of the annual rings. Well-formed concentric growth rings were found for both of the sites from the Rio Paru, and a new chronology dating from 1759–2016 was developed based on 120 radii from 60 trees at RPB using graphical comparisons of ring width time series and visual dating under the microscope (Douglass, 1941; Stokes & Smiley, 1996). All cross sections from the two sites on the Rio Paru can be dated with dendrochronology, but some specimens were too short or irregular to have value for long chronology development. The chronology from RPB used for these analyses was instead based on the most strongly cross-correlated subset of 50 dated radii from 22 trees. The "Schulman Shift" (Schulman, 1956) was not used in the dating of the ring width series from Rio Paru sites because the wet season typically begins in February in this equatorial area

and both chronologies respond to precipitation during the same calendar year as the wet season (Granato-Souza et al., 2018).

The dendrochronologically dated ring widths were measured with a precision of 0.01 mm on a stage micrometer. The computer program COFECHA (Holmes, 1983) was used to check dating and measurement accuracy. The dated time series were detrended and standardized using data adaptive cubic smoothing splines with a 50% frequency response of 100 years to minimize short-period growth excursions associated with nonclimatic stand dynamics in these closed canopy rain forests using the ARSTAN program (Cook & Krusic, 2005). All ring width index series were averaged on a year-by-year basis producing the final robust mean index chronology (Cook & Krusic, 2005).

Correlation and composite analyses with monthly precipitation data were computed in order to define the strength and seasonality of the rainfall signal of this new *Cedrela* chronology. Monthly precipitation data for all Brazilian recording stations in the Amazon basin were obtained from the Agência Nacional de Águas from 1920 to 2016 (ANA, National Water Agency). The RPB chronology was correlated with the gridded monthly precipitation data from the Global Precipitation Climatology Centre (GPCC) V7 0.5° data set (Becker et al., 2013; Schneider et al., 2014) to identify the strongest regional rainfall response. The GPCC data for the wet season (February–July) were then regionally averaged, and that instrumental time series was used for reconstruction.

A reconstruction based on the single new RPB chronology was computed using bivariate regression to calibrate the standardized ring width chronology from the Rio Paru with February–July precipitation data extracted from the 0.5° gridded GPCC data set for 0–1°S and 56–57°W near Santarem and 150 km west of the remote RPB collection site. The tree-ring chronology was calibrated with wet season rainfall totals from 1978 to 2016 when the GPCC precipitation data from the eastern Amazon are based on several reporting stations and appear to be most homogeneous. Validation of the reconstruction prior to the calibration interval is difficult because rainfall stations in the region become increasingly scarce. Nonetheless, the reconstruction was compared with the regional average GPCC wet season rainfall data and with the totals recorded at the Santarem station from 1939–1977. The derived reconstruction was also evaluated during the calibration interval using the leave-one-out cross-validation reduction of error (CVRE) statistic, a regression diagnostic similar to the Prediction Error Sum of Squares (Allen, 1974).

Semiparametric 90% prediction intervals were computed for each year of the reconstruction to estimate the uncertainty associated with the reconstructions through time (Cook et al., 2013) using the point-by-point regression program of Cook et al. (1999); E.R. Cook, Personal Communication, 2019). These prediction intervals are based on standard least squares theory (Olive, 2007; Seber & Lee, 2003) and the maximum entropy bootstrap method (Vinod, 2006) to randomly rearrange the predictors and predictands. In the generation of the semiparametric prediction intervals, estimates of regression model interpolations and extrapolations were generated from the Hat matrix of predictors and were used to identify extrapolated values likely to be least reliable for interpretation in the reconstruction (Wiesberg, 1985).

Singular spectrum analysis (Ghil et al., 2002; St. George & Ault, 2011) was performed to identify important frequency components in the 258-year reconstruction. The Mann-Kendall test (Kendall, 1975; Mann, 1945) was used assess the presence of significant (p < 0.05) monotonic trends in the derived reconstructed time series of wet season precipitation. Spectral and cross spectral coherence analyses were used to compare the two Rio Paru tree-ring chronologies in the frequency domain (Percival & Constantine, 2006), and the 95% confidence level was estimated through 10,000 runs of Monte Carlo simulations.

3. Results and Discussion

Instrumental precipitation measurements have become widespread in the Amazon basin in only the last 70 years (Figure 1). Just 33 stations were operating in the basin in 1940, but 467 were reporting observations in 2000 (Figures 1a– 1d). However, many station observations have not been continuous even in recent decades. The precipitation records at Rio Branco in the southwest and Manaus in the central Amazon are the two longest in the Brazilian Amazon. Both of these important records are subject to significant uncertainties during the earliest years of observation. The wet and dry season totals for Manaus are presented in Figures 1e and 1f, and both are inhomogeneous when cumulatively plotted with the long river level record of the Rio





Figure 2. Average anomalies in wet season (February–July) precipitation are plotted for all available instrumental stations in and near the Amazon basin of Brazil for the five wettest (a) and five driest years (b) recorded at Manaus from 1980–2016 (circles = positive, triangles = negative anomalies). February–July precipitation totals are also plotted for all stations during the five highest (c) and five lowest years of tree growth from 1980–2016 (d) in the new Rio Paru-Baixo tree-ring chronology (location = $0.997522^{\circ}S$ 53.2667°W). The number of above and below average (below/above) stations is listed below each map.

Negro at Manaus in double mass analyses (Kohler, 1949). The wet season totals appear to be problematic before circa 1920 and the dry season totals before circa 1960 (Figures 1e and 1f). Both seasonal totals exhibit positive trend during the early 20th century that is not strongly present in the maxima, minima, or annual Rio Negro or Amazon River level data (e.g., Gloor et al., 2013; Labat et al., 2004). The early precipitation data at Manaus may be impacted by irregularities in instrumentation or observation that would complicate the detection of low-frequency variability. Moisture sensitive tree-ring chronologies may therefore provide a valuable proxy perspective on decadal to multidecadal variability both during and preceding the instrumental period.

The tree-ring chronology developed at the RPB site is based on highly polished cross sections of *C. odorata* that exhibit obvious and well-defined annual growth rings that are correctly dated to the calendar year of formation. The Rio Paru *Cedrela* is fortunately not subject to the kind of obscure, suppressed, and often locally absent growth rings observed on many tropical tree species. The internal crossdating among the individual trees and radii at RPB is excellent, which is quantified by the fact that the Expressed Population Signal statistic (EPS; Wigley et al., 1984) never falls below the 0.85 threshold from 1759–2016 (ranging from 0.87 to 0.97 for 50-year segments) and the average correlation among all time series (RBAR) ranges from 0.27 to 0.51.

The development of the RPB chronology confirms the exact dating of the original master chronology of *Cedrela* from the nearby RPA site. The two mean index chronologies are correlated over their 230-year common interval at r = 0.63 (p < 0.0001; 1786–2016) using the autoregressively modeled "white noise" residual versions for both chronologies. This between site correlation increases to r = 0.72 (p < 0.0001) for the period from 1851–2016 when both chronologies are more fully replicated. The decadal components of both chronologies indicate similar periods of above and below average tree growth over the past 230 years, likely in





Figure 3. (a) Instrumental GPCC (dashed) and tree-ring reconstructed (solid) wet season precipitation totals (February–July) for the eastern equatorial Amazon are plotted from 1978–2016 (RSQ = R^2 adj). (b) The full reconstruction of wet season precipitation for the eastern Amazon is plotted from 1759–2016 (black) along with the 90% prediction intervals (light red). The leading SSA waveform identified in the reconstruction is also plotted (smooth black), highlighting the strong multidecadal variability in the reconstructed wet season precipitation over the eastern Amazon. (c) The 10 driest years in the wet season precipitation reconstruction for the eastern Amazon are used to map SST anomalies for the period from 1856–2016 (SSTs from Kaplan et al., 1998; grid point means significantly different from all remaining years are stippled, p < 0.05).

response to multiyear droughts and pluvials. Cross-spectral analysis indicates that the two chronologies are coherent at periods from 2–32 years at p < 0.05 and 2–51 years at p < 0.10.

The new *C. odorata* chronology from RPB not only confirms the original dating of our first *Cedrela* chronology but also is more strongly coupled with regional rainfall totals and large-scale ocean-atmospheric variability. Station observations of wet season rainfall from ANA, Brasil (2019) were averaged for the five wettest and five driest February–July precipitation totals at the Manaus station from 1980–2016 to illustrate this coherence of precipitation extremes over the Amazon basin, particularly the dry extremes (Figure 2ab; note that all 5 years had to be available for a station to be included in any one of the four maps in Figure 2). When the five highest and five lowest values in the RPB ring width chronology were used to composite the station

precipitation data, a consistent basin-wide pattern of above and below average wet season rainfall was also observed (Figures 2c and 2d), very similar to the precipitation anomalies associated with extremes in the Manaus gage. Similar results were obtained using the February–July precipitation extremes recorded at the Santarem station and in the gridded GPCC monthly precipitation data (also for February–July; not shown). Just 5 years were used in this analysis from 1980–2016, but the association between extremes in the RPB chronology and station rainfall totals is present when compared with the many fewer stations available earlier in the 20th century in the ANA compilation (not shown). Correlation analyses comparing the RPB chronology with February–July precipitation in the gridded GPCC data set identify significant (p < 0.05) correlations over the central and eastern Amazon basin for 1978–2016, similar to, though less widespread than, the composite pattern illustrated in Figure 2d.

The single RPB chronology was calibrated with the regional GPCC February-July precipitation data (0-1°S and 56-57°W) using bivariate regression and explains 42% of wet season precipitation variability from 1978-2016 (R²adj = 0.42, adjusted for the loss of degrees of freedom; Figures 3a and b). The February-July interval is the wet season for the Rio Paru region of the northeast Amazon. The region of GPCC data used for calibration with the RPB chronology was selected because it is well correlated with the RPB chronology from 1978-2016 (including the GPCC data from the remote Rio Paru basin in this regional average did not improve the calibration). The cross-validation reduction of error statistic indicates little loss of explained variance when tested against withheld instrumental observations during the calibration period (CVRE = 0.39). However, the reconstruction is not significantly correlated with the extracted GPCC data from 1939–1977 (r = 0.21). This lack of correlation during the precalibration period may be due in part to deficiencies in the gridded instrumental data. The mean and variance of the regional GPCC precipitation data become increasingly impacted by the lack of stations observations before 1978, limiting the time interval available for cross-validation of the reconstruction. The reconstruction is modestly correlated from 1939-1977 with February-July precipitation recorded at Santarem 250 km west of the collection site (r = 0.38, p < 0.05, r = 0.46 omitting 1943). The reconstruction is also significantly correlated from 1939– 1977 with a four-station instrumental average of February-November precipitation for the central and eastern Amazon (Granato-Souza et al., 2018) at r = 0.41 (p < 0.01). The association between reconstructed drought years in the eastern Amazon with warm sea surface temperature (SST) anomalies in the tropical Pacific and Atlantic from 1870-2016 (Figure 3c) provides some additional validation of the reconstruction.

The reconstruction is plotted from 1759–2016 in Figure 3b along with the 90% regression prediction intervals calculated from a suite of pseudoreconstructions (Cook et al., 2013). The reconstruction exhibits strong interannual and low-frequency variability, including decadal to multidecadal episodes of drought and wetness. The mid-19th century drought is especially notable because it extended with little relief from 1864–1882 (Figure 3b) and was also reconstructed by Granato-Souza et al. (2018). The two reconstructions now available from the Rio Paru sector therefore indicate that severe and sustained drought has been a feature of natural climate variability over the eastern equatorial Amazon, including decadal moisture regimes that exceeded any recorded during the relatively short instrumental period (Figure 3).

Spectral analysis identifies concentrations of reconstructed precipitation variance at periods of approximately 4.0, 6.0, and 11.0, and especially at 35 years (p < 0.05, using the Blackman-Tukey method). Singular spectrum analysis indicates that multidecadal variability represents approximately 20% of the variance in reconstructed wet season precipitation totals over the eastern Amazon, highlighted by the leading 35.4-year waveform in Figure 3b (bandwidth = 20). Similar and significant multidecadal power peaking at 33.4 years (p < 0.01) is detected in the instrumental observations of February–July rainfall totals at Manaus from 1901–2016, the longest instrumental precipitation record from the central and eastern Amazon. The homogeneity of the Manaus precipitation measurements during the early 20th century is doubtful, but significant multidecadal power is nevertheless present in February–July rainfall when the analyses are restricted to the period from 1920–2016 (p < 0.01). The Rio Negro maximum annual stream level record at Manaus also has significant multidecadal spectral power from 1920–2013 (p < 0.05), but not for the full period of observation from 1903–2016 (e.g., Labat et al., 2004).

The multidecadal variability of reconstructed wet season precipitation in the eastern Amazon could have practical implications for interpretation of trends in the short instrumental precipitation data from the region. The reconstruction was tested for significant monotonic trends using the Mann-Kendal test applied to 30-year

segments and a 1-year time step proceeding through the time series. Significant (p < 0.05) positive and negative trends of 30-year duration were both detected in the reconstruction during the instrumental and preinstrumental period. If multidecadal swings are an important component of natural wet season variability, then significant linear trends of 30-year duration are likely to also be present. The extended perspective provided by this reconstruction suggests that the recent decade scale changes in hydroclimate over the Amazon may not be unprecedented over the past 258 years. The wet season precipitation estimates provide analogs of both extreme individual years (such as 1818 and 1865) and low-frequency changes in the mean.

The wet season reconstruction is only weakly correlated with SSTs in the tropical Pacific (not shown), but composite mapping of the 10 driest years in the reconstruction over the period from 1856–2016 indicates a strong negative response to El Niño conditions in the tropical Pacific and a weaker but still significant negative response to SSTs in the tropical North Atlantic (Figure 3c). This pattern of response is similar to the SST forcing of instrumental precipitation variability over the Amazon basin (Enfield, 1996; Dettinger et al., 2001; Yoon & Zeng, 2010; Fernandes et al., 2011; Yeh et al., 2018). Unlike the instrumental precipitation data, the tree-ring estimates are not significantly associated with cold SSTs in the tropical Pacific SST anomalies that could be consistent with the Pacific Decadal Oscillation (Mantua et al., 1997) or more broadly the Inter-Decadal Pacific Oscillation (IPO; Dong and Dai, 2014). The quasi-oscillatory IPO has full period of 40–60 years, with phase reversals every 20–30 years (Dong & Dai, 2015) and may have influenced the low-frequency variability of reconstructed precipitation in the eastern Amazon over the past 258 years. The reconstruction is only marginally correlated with the Atlantic Multi-Decadal Oscillation (r = 0.159; p < 0.05; June and July average, 1856–2016).

Multidecadal variability has important implications for forest ecosystem dynamics and biogeochemical cycling in Amazonia. The recently observed decline in the long-term carbon sink of Amazon rainforests (Brienen et al., 2015) has been attributed to increasing climate variability, including the severe droughts of 2005 and 2010 (Aragao et al., 2018; Lewis et al., 2011). However, the influence of multidecadal variability on rain forest dynamics in the Amazon is difficult to investigate because direct forest monitoring data rarely exceed 30 years of record.

4. Conclusions

The new tree-ring chronology from RPB provides important validation of the master dating chronology for *C. odorata* in the eastern Amazon and confirms the climate response first identified by Granato-Souza et al. (2018). Tree-ring chronology development and climate reconstruction in the Amazon basin are both predicted on the consistent formation of annual growth rings and their exact calendar year dating with dendrochronology. The evidence for the exact dating of *C. odorata* ring width data from the Rio Paru now includes the significant correlation between the two chronologies from the Rio Paru, the similar and significant response of both chronologies to instrumental precipitation totals, and the link between reconstructed drought extremes and SST anomalies in the tropical Pacific where ENSO conditions have a strong impact on instrumental precipitation over the eastern Amazon.

The new Cedrela chronology has been used to reconstruct wet season rainfall totals over the eastern equatorial Amazon from 1759–2016. This reconstruction confirms the severe sustained drought in the mid-19th century first estimated with the RPA chronology by Granato-Souza et al. (2018). This mid-19th century drought is registered by both *Cedrela* chronologies from the Rio Paru basin and in both cases is estimated to have been more extreme than any reconstructed during the relatively short instrumental data for the eastern Amazon. Another intense and prolonged drought is reconstructed during the late-18th century (Figure 3b), which if confirmed with additional tree-ring data would suggest that decadal droughts may be underestimated in the available instrumental precipitation observations for the region.

Tree-ring reconstructed wet season precipitation totals in the eastern Amazon have been subject multidecadal regimes over the past 258 years. This multidecadal variability of reconstructed February–July rainfall has impacted the frequency of drought extremes in the eastern Amazon. Several decadal to multidecadal regimes without droughts in the lowest 10th percentile are reconstructed from 1759–2016 (Figure 3b). Similar decadal episodes without drought or flood extremes have been noted in other instrumental precipitation and river level data from the Amazon basin. If quasiperiodic low-frequency variability is truly an



important component of natural precipitation that can impact the occurrence of drought and wetness extremes, then it needs to be validated because it might help explain some of the recent increase in floods and drought witnessed in the Amazon and would likely interact with predicted anthropogenic forcing of future precipitation extremes.

Acknowledgments

Research funding was provided by the U.S. National Science Foundation (Grant AGS-1501321). D. Granato-Souza was funded in Brazil by a grant from the Coordination for the Improvement of Higher Education Personnel (CAPES) to the Federal University at Lavras. We gratefully acknowledge the extensive logistical support and wood donations from the CEMAL and RRX Florestal logging firms and Sr. Evandro Dalmaso, Sra-Eliane Dalmaso, Gilberto Oliveira Santos, Valcemir Oliveira de Aviz, Walk Ferreira Santos, Marcos Gonçalves Durval, Edson Sousa de Araújo, and Rodrigo Montezano. We also thank Sr. Robson Azeredo, Adelson da Luz Oliveira, Claudio, Jailton, Chico, Félix, Mauro Caldas. We acknowledge the data provided by the Agência Nacional de Águas, the Climatic Research Unit, University of East Anglia, and the use of the KNMI Climate Explorer. The data described in this article have been contributed to the International Tree-Ring Data Bank at the NOAA Paleoclimatology Program (https://www.ncdc.noaa.gov/data-access/paleoclimatology-data/datasets/ tree-ring).

References

- Agencia Nacional de Aguas-ANA, Brasil (2019), Sistema Nacional de Informacoes sobre Recursos Hidricos SNIRH. Modulo Hidroweb. Brasilia: ANA, 2019. Disponivel em: http://www.snirh.gov.br/hidroweb/ Accessed: May 23, 2019.
- Allen, D. M. (1974). The relationship between variable selection and data augmentation and a method for prediction. *Technometrics*, *16*, 125–127.
- Aragao, L. E. O. C., Anderson, L. O., Fonseca, M. G., et al. (2018). 21st Century drought-related fires counteract the decline of Amazon deforestation carbon emissions. *Nature Communications*, 9(1), 536. https://doi.org/10.1038/s41467-017-02771-y
- Ault, T. R., Cole, J. E., & George, S. S. (2012). The amplitude of decadal to multidecadal variability in precipitation simulated by state-of-the-art climate models. *Geophysical Research Letters*, 39, L21705. https://doi.org/10.1029/2012GL053424
- Barichivich, J., Gloor, M., Peylin, P., Brienen, R. J. W., Schöngart, J., Espinoza, J. C., & Pattnayak, K. C. (2018). Recent intensification of Amazon flooding extremes driven by strengthened Walker Circulation. *Science Advances*, 2018(4), eaat8785.
- Becker, A., Finger, P., Meyer-Christoffer, A., Rudolf, B., Schamm, K., Schneider, U., & Ziese, M. (2013). A description of the global land-surface precipitation data products of the Global Precipitation Climatology Centre with sample applications including centennial (trend) analysis from 1901-present. *Earth System Science Data*. https://doi.org/10.5194/essd-5-71-2013
- Brienen, R. J. W., Helle, G., Pons, T. L., Guyot, J. L., & Gloor, M. (2012). Oxygen isotopes in tree rings are a good proxy for Amazonian precipitation and ENSO variability. Proceedings of the National Academy of Sciences, 109, 16957–16962.
- Brienen, R. J. W., Phillips, O., Feldpausch, T., et al. (2015). Long-term decline of the Amazon carbon sink. Nature, 519, 344.
- Chaudhari, S., Pokhrel, Y., Moran, E., & Miguez-Macho, G. (2019). Multi-decadal hydrologic change and variability in the Amazon River basin: understanding terrestrial water storage variations and drought characteristics. *Hydrology and Earth System Sciences*, 23, 2841–2862.
- Chelliah, M., & Bell, G. D. (2004). Tropical multidecadal and interannual climate variability in the NCEP–NCAR Reanalysis. Journal of Climate, 17(9), 1777–1803. https://doi.org/10.1175/1520-0442(2004)017<1777:TMAICV>2.0.CO;2
- Cook, E.R., and P.J. Krusic (2005) Program ARSTAN: A tree-ring standardization program based on detrending and autoregressive time series modeling, with interactive graphics. Manuscript on file, Tree-Ring Lab, Lamont Doherty Earth Observatory of Columbia University, Palisades, NY.
- Cook, E. R., Meko, D. M., Stahle, D. W., & Cleaveland, M. K. (1999). Drought reconstructions for the continental United States. Journal of Climate, 12, 1145–1162.
- Cook, E. R., Palmer, J. G., Ahmed, M., Woodhouse, C. A., Fenwick, P., Zafar, M. U., et al. (2013). Five centuries of Upper Indus River flow from tree ring. *Journal of Hydrology*, 486, 365–375.
- Dettinger, M. D., Cayan, D. R., McCabe, G. M., & Marengo, J. A. (2001). Multiscale streamflow variability associated with El Niño/Southern Oscillation. In H. F. Diaz, & V. Markgraf (Eds.), El Niño and the Southern Oscillation-Multiscale variability and global and regional impacts, (pp. 113–146). Cambridge: Cambridge University Press.
- Dong, B., & Dai, A. (2015). The influence of the Interdecadal Pacific Oscillation on temperature and precipitation over the globe. *Climate Dynamics*. https://doi.org/10.1007/s00382-015-2500-x
- Douglass, A. E. (1941). Crossdating in dendrochronology. Journal of Forestry, 39, 825-831.
- Enfield, D. B. (1996). Relationships of inter-American rainfall to tropical Atlantic and Pacific SST variability. *Geophysical Research Letters*, 23, 3305–3308.
- Espinoza, J. C., Marengo, J. A., Ronchail, J., Carpio, J. M., Flores, L. N., & Guyot, J. L. (2014). The extreme 2014 flood in south-western Amazon basin: The role of tropical-subtropical South Atlantic SST gradient. *Environmental Research Letters*, 9, 124007.
- Feng, S., & Fu, Q. (2013). Expansion of global drylands under a warming climate. Atmospheric Chemistry and Physics, 13, 10081–10094.
 Fernandes, K., Baethgen, W., Bernardes, S., DeFries, R., DeWitt, D. G., Goddard, L., et al. (2011). North Tropical Atlantic influence on western Amazon fire season variability. *Geophysical Research Letters*, 38. https://doi.org/10.1029/2011GL047392
- Fernandes, K., Giannini, A., Verchot, L., Baethgen, W., & Pinedo-Vasquez, M. (2015). Decadal covariability of Atlantic SSTs and western Amazon dry-season hydroclimate in observations and CMIP5 simulations. *Geophysical Research Letters*, 42, 6793–6801. https://doi.org/ 10.1002/2015GL063911
- Garreaud, R. D., Vuille, M., & Compagnucci, R. (2009). Present-day South American climate. Paleogeography, Paleoclimatology, Paleoecology, 281, 180–195.
- Ghil, M., Allen, M. R., Dettinger, M. D., Ide, K., Kondrashov, D., Mann, M. E., et al. (2002). Advanced spectral methods for climatic time series. *Reviews of Geophysics*, 40(1), 1003. https://doi.org/10.1029/2000RG000092
- Gloor, M., Brienen, R. J. W., Galbraith, D., Feldpausch, T. R., Schöngart, J., Guyot, J. L., et al. (2013). Intensification of the Amazon hydrological cycle over the last two decades. *Geophysical Research Letters*, 40, 1729–1733. https://doi.org/10.1002/grl.50377
- Granato-Souza, D., Stahle, D. W., Barbosa, A. C., Feng, S., Torbenson, M. C. A., Pereira, G., et al. (2018). Tree rings and rainfall in the equatorial Amazon. *Climate Dynamics*. https://doi.org/10.1007/s00382-018-4227-y
- Holmes, R. L. (1983). Computer-assisted quality control in tree-ring dating and measurement. Tree-Ring Bulletin, 43, 69-78.
- Kaplan, A., Cane, M. A., Kushnir, Y., Clement, A. C., Benno Blumenthal, M., & Rajagopalan, B. (1998). Analyses of global sea surface temperature 1856–1991. Journal of Geophysical Research, 103, 18,567–18,589.
- Kendall, M. G. (1975). Rank correlation methods, (4th ed.). London: Charles Griffin.
- Khanna, J., Medvigy, D., Fueglistaler, S., & Walko, R. (2017). Regional dry-season climate changes due to three decades of Amazonian deforestation. *Nature Climate Change*, 7, 200–204.
- Kohler, M. (1949). On the use of double-mass analysis for testing the consistency of meteorological records and for making required adjustments. *Bulletin of the American Meteorological Society*, *30*, 188–189.
- Labat, D., Ronchail, J., Callede, J., Guyot, J. L., Oliveira, E., & Guimaraes, W. (2004). Wavelet analysis of Amazon hydrological regime variability. *Geophysical Research Letters*, 31, L02501. https://doi.org/10.1029/2003GL018741



Lewis, S. L., Brando, P. M., Phillips, O. L., van der Heijden, G. M. F., & Nepstad, D. (2011). The 2010 Amazon Drought. Science, 331. https://doi.org/10.1126/science.1200807

Mann, H. B. (1945). Non-parametric tests against trend. Econometrica: Journal of Econometric Society, 13, 245-259.

Mantua, N. J., Hare, S. R., Zhang, Y., Wallace, J. M., & Francis, R. C. (1997). A Pacific interdecadal climate oscillation with impacts on salmon production. *Bulletin of the American Meteorological Society*, 78, 1069–1079.

Marengo, J. A. (2004). Interdecadal variability and trends of rainfall across the Amazon basin. *Theoretical and Applied Climatology*, 78, 79–96.

Marengo, J. A., & Espinoza, J. C. (2016). Extreme seasonal droughts and floods in Amazonia: causes, trends and impacts. *International Journal of Climatology*, 36. https://doi.org/10.1002/joc.4420

Nobre, C. A., Sampiao, G., Borma, L. S., Castilla-Rubio, J. C., Silva, J. S., & Cardoso, M. (2016). Land-use and climate change risks in the Amazon and the need of a novel sustainable development paradigm. *Proceedings of the National Academy of Sciences*, 113, 10759–10768.

Olive, D. J. (2007). Prediction intervals for regression models. *Computational Statistics and Data Analysis*, *51*, 3115–3122. Percival, D. B., & Constantine, W. L. B. (2006). Exact simulation of Gaussian time series from nonparametric spectral estimates with

application to bootstrapping. Statistics and Computing, 16, 25-35. https://doi.org/10.1007/s11222-006-5198-0

Ronchail, J., Cochonneau, G., Molinier, M., Guyot, J. L., Miranda Chaves, A. G., Guimaraes, V., & Oliveira, E. (2002). Interannual rainfall variability in the Amazon basin and sea surface temperatures in the equatorial Pacific and the tropicl Atlantic Oceans. *International Journal of Climatology*, 22, 1663–1686.

Ropelewski, C. F., & Halpert, M. S. (1987). Global and regional scale precipitation patterns associated with the El-Niño Southern Oscillation. *Monthly Weather Review*, 115, 1606–1626.

Schneider, U., Becker, A., Finger, P., Meyer-Christoffer, A., Ziese, M., & Rudolf, B. (2014). GPCC's new land surface precipitation climatology based on quality-controlled in situ data and its role in quantifying the global water cycle. *Theoretical and Applied Climatology*. https://doi.org/10.1007/s00704-013-0860-x

Schulman, E. (1956). Dendroclimatic changes in semiarid America, (p. 142). Tucson: University of Arizona Press.

Seber, G. A. F., & Lee, A. J. (2003). Linear regression analysis, (2nd ed.). New York: Wiley-Interscience.

St. George, S., & Ault, T. R. (2011). Is energetic decadal variability a stable feature of the central Pacific Coast's winter climate? Journal of Geophysical Research, 116. https://doi.org/10.1029/2010JD015325

Stokes, M. A., & Smiley, T. L. (1996). An introduction to tree-ring dating, (p. 73). Tucson: University of Arizona Press.

Vinod, H. D. (2006). Maximum entropy ensembles for time series inference in economics. Journal of Asian Economics, 17, 955–978. https:// doi.org/10.1016/j.asieco.2006.09.001

Wang, G., Sun, S. & Mei, R. (2011) Vegetation dynamics contributes to the multi-decadal variability of precipitation in the Amazon region. Geophysical Research Letters, 38, L19703.

Wang, X., Edwards, R. L., Auler, A. S., Cheng, H., Kong, X., Wang, Y., et al. (2017). Hydroclimate changes across the Amazon lowlands over the past 45,000 years. *Nature*, 541(7636), 204–207.

Wiesberg, S. (1985). Applied Linear Regression, (2nd ed.). Hoboken, NJ: Wiley.

Wigley, T. M. L., Briffa, K. R., & Jones, P. D. (1984). On the average value of correlated time series, with applications in dendroclimatology and hydrometeorology. *Journal of Climate and Applied Meteorology*, 23, 201–213. https://doi.org/10.1175/1520-0450(1984)023<0201: OTAVOC>2.0.CO;2

Yeh, S. W., Cai, W., Min, S. K., McPhaden, M. J., Dommenget, D., Dewitte, B., et al. (2018). ENSO atmospheric teleconnections and their response to greenhouse gas forcing. *Reviews of Geophysics*, 56, 185–206.

Yoon, J. H., & Zeng, N. (2010). An Atlantic influence on Amazon rainfall. *Climate Dynamics*, 34, 249–264. https://doi.org/10.1007/ s00382-009-0551-6

Zeng, N. (1999). Seasonal cycle and interannual variability in the Amazon hydrologic cycle. Journal of Geophysical Research, 104(D8), 9097–9106. https://doi.org/10.1029/1998JD200088